Author’s response to reviews

Title: Risk Factors of Morbidity among Children under Age Five in Ethiopia

Authors:
Kasahun Geneti (kastake10@gmail.com)
Temesgen Zewotir (zewotir@ukzn.ac.za)
Denis Ndanguza (dndanguza@gmail.com)

Version: 1 Date: 08 Feb 2019

Author's response to reviews:

Manuscript is revised as per the comments. The manuscript is revised as per the format of required guidelines of BMC Public Health. Some of statistical descriptions are also removed.